
\documentstyle[12pt]{article}
\textwidth 15true cm
\textheight 8.65true in
\oddsidemargin 0.25true in
\evensidemargin 0.25true in
\topmargin -0.3true in
\headsep 0.4true in

\begin{document}

{\bf SCI199Y: Short project 1} \hfill {\it due October 31: extension to
November 7 okay}

\bigskip

\begin{itemize}
\item Write a review of one chapter of {\it VDQI} (your choice).
Summarize the main ideas of the chapter briefly. Discuss what you
think are the strongest and weakest points Tufte makes in the chapter.
Comment on how well the accompanying graphics help him make his points.
Did you like the chapter?  Did you like the book?

(Your review should be between one and two typed pages long.)
\end{itemize}
\end{document}


 
